

REVIEW HISTORY

Paper: **“Equalizing Political Participation and, in turn, Political Influence with Civil Liability Rules”**

Corresponding Author: Sigmund A. Horvitz
Email: shirlvitz@aol.com

Doi: 10.19044/elv.v7no1a1

Peer review:

Reviewer 1: Emilia Alaverdov
Georgian Technical University, Georgia

Reviewer 2: Blinded

Published: 31.03.2020

